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ABSTRACT 

The design, performance and safety analysis of nuclear reactors, in particular, Boiling 

Water Reactors (BWRs), require accurate and extensive information about the vapor-liquid 

water flow dynamics. Computational fluid dynamics (CFD) has evolved over the last several 

decades as an important tool for improving the safety and performance of the nuclear reactors. 

However, CFD results must be validated and verified by reliable experimental data (at multiple 

scales) to guarantee, with adequate confidence, the reliability and worth of the model 

predictions. Most of the experimental data available in the literature suffer from the limitations 

posed by conventional invasive techniques, which interfere with the flow itself. Their 

limitations are magnified in the context of flow boiling wherein bubble nucleation and vapor 

evolution is known to be a complex and multi-scale problem. Additionally, the experiments 

performed in the past are predominantly based on test sections which are single rod annular 

geometry or rectangular geometry. Also, most of the experimental data available in the past are 

based on systems operating at a very high pressure and high flow rates. 

The present thesis focuses on investigating multiphase flow dynamics of convective 

boiling flows on two test sections. In this process, two non-invasive techniques i.e. γ-ray 

densitometry and Radioactive Particle Tracking (RPT) has been developed for performing 

experiments pertaining to boiling flows. The first test section is a vertical annular channel of 

single heater rod whereas the second experimental test section has the provision of holding 

different heater rods arrangements. For both experimental test sections, experiments were 

performed at the different inlet flow conditions, i.e., inlet flow rates and inlet liquid 

temperatures. In the case of the second experimental test section, experiments were also 

performed at various heater rods configuration. Radial and axial void fraction measurements 

were performed on these experimental test section using non-invasive γ-ray densitometry and 

Radioactive Particle Tracking (RPT), respectively. Non-invasive Radioactive Particle Tracking 
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(RPT) has been used for measuring liquid velocity field, mean flow patterns and “turbulent” 

kinetic energy. The unique aspect of this thesis lies in the fact that non-invasive radiation based 

techniques have been used for all experimental measurements. 

Another aspect of the present thesis was to make RPT suitable for systems having large 

interrogation volumes and possessing internals such as in the present case. Two new 

reconstruction algorithms were developed in this thesis. First was the extension of classical 

Monte-Carlo (MC) technique known as “MC-RGA” technique, whereby the convergence and 

computational speed of the algorithm are enhanced using Real Coded Genetic Algorithm 

(RGA). This modification results in drastic reduction in computational time required for 

detector parameter estimation and altogether eliminates the need for the “distance-count map,” 

which was earlier inherent to RPT experimentation. RPT application was limited to convective 

boiling flow experiments owing to the “shadow effect” caused by the solid heater rods. The 

“shadow effect” makes classical position reconstruction algorithm based on Monte Carlo (MC) 

technique futile. A novel tracer particle reconstruction algorithm based Support Vector 

Regression (SVR) and Relevance Vector Machine (RVM) has been applied in this study. The 

machine learning based reconstruction algorithms were very fast and accurate in terms of 

position reconstruction. 

Apart from providing mean phase velocities and other “turbulence” quantities, RPT also 

provides the actual Lagrangian trajectory of the tracer particle mimicking liquid phase in the 

present study. Hence, it is an appropriate method for the detailed characterization of the 

complex liquid dynamics observed in boiling two-phase flows. The position time series of the 

radioactive tracer particle conveying information on the liquid motion were examined by 

Recurrence Plots (RP) and Recurrence Quantification Analysis (RQA). The Lagrangian 

trajectories were also quantified using mean displacement model. Eddy diffusivity was 

calculated in axial and radial direction for different experimental conditions.  



सार 

परमाणु ररएक्टरों के डिजाइन, प्रदर्शन और सुरक्षा डिश्लेषण, डिर्ेष रूप से, उबलत ेजल 

ररएक्टरों (बीिब्लूआर), को िाष्प-तरल जल प्रिाह गडतर्ीलता के बारे में सटीक और व्यापक 

जानकारी की आिश्यकता होती ह।ै परमाणु ररएक्टरों की सुरक्षा और प्रदर्शन में सुधार के डलए 

डपछले कई दर्कों में कम्प्यूटेर्नल तरल गडतर्ीलता (सीएफिी) डिकडसत हुआ ह।ै हालाांकक, 

सीएफिी पररणामों को डिश्वसनीय प्रयोगात्मक िेटा (एकाडधक पैमाने पर) द्वारा पयाशप्त 

आत्मडिश्वास, डिश्वसनीयता और मॉिल भडिष्यिाडणयों के मूल्य के साथ गारांटी के डलए सत्याडपत 

और सत्याडपत ककया जाना चाडहए। साडहत्य में उपलब्ध अडधकाांर् प्रयोगात्मक िेटा परांपरागत 

आक्रामक तकनीकों द्वारा उत्पन्न सीमाओं से ग्रस्त हैं, जो प्रिाह के साथ हस्तके्षप करते हैं। उनकी 

सीमाओं को प्रिाह उबलत ेसांदभश में बढाया जाता ह ैडजसमें बबल न्यूडक्लयरे्न और िाष्प डिकास 

को जरटल और बहु-पैमाने की समस्या माना जाता ह।ै इसके अडतररक्त, अतीत में ककए गए 

प्रयोग मुख्य रूप से परीक्षण खांिों पर आधाररत होते हैं जो ससांगल रॉि क्युलर ज्याडमडत या 

आयताकार ज्याडमडत होत ेहैं। इसके अलािा, अतीत में उपलब्ध अडधकाांर् प्रयोगात्मक िेटा बहुत 

उच्च दबाि और उच्च प्रिाह दर पर चल रह ेडसस्टम पर आधाररत होते हैं। 

ितशमान थीडसस दो परीक्षण खांिों पर सांिहनी उबलत ेप्रिाह की मल्टीफेस प्रिाह गडतर्ीलता की 

जाांच पर कें कित ह।ै इस प्रकक्रया में, दो गैर-आक्रामक तकनीकें यानी γ-ray densitometry 

और रेडियोधमी कण टै्रककां ग (आरपीटी) उबलत ेप्रिाह से सांबांडधत प्रयोग करन ेके डलए डिकडसत 

की गई ह।ै पहला परीक्षण खांि एकल हीटर रॉि का एक लांबित कडणका चैनल ह ैजबकक दसूरे 

प्रयोगात्मक परीक्षण खांि में डिडभन्न हीटर रॉि व्यिस्था आयोडजत करने का प्रािधान ह।ै 

प्रयोगात्मक परीक्षण खांि दोनों के डलए, डिडभन्न इनलेट प्रिाह डस्थडतयों, यानी इनलेट प्रिाह 

दर और इनलेट तरल तापमान पर प्रयोग ककए जाते थे। दसूरे प्रयोगात्मक परीक्षण खांि के मामल े

में, डिडभन्न हीटर रॉि डिन्यास पर प्रयोग भी ककए गए थे। क्रमर्ः गैर-आक्रामक γ-रे 

densitometry और रेडियोधमी कण टै्रककां ग (आरपीटी) का उपयोग कर इन प्रयोगात्मक 

परीक्षण खांि पर रेडियल और अक्षीय र्ून्य अांर् माप ककया गया था। गैर-आक्रामक रेडियोधमी 

कण टै्रककां ग (आरपीटी) का उपयोग तरल िेग क्षेत्र, माध्य प्रिाह पैटनश और "अर्ाांत" गडतर्ील 

ऊजाश को मापन ेके डलए ककया गया ह।ै इस थीडसस का अनूठा पहल ूइस तथ्य में डनडहत ह ैकक 

सभी प्रयोगात्मक मापों के डलए गैर-आक्रामक डिककरण आधाररत तकनीकों का उपयोग ककया 

गया ह।ै 

ितशमान थीडसस का एक अन्य पहल ूआरपीटी को बड ेपूछताछ िाल ेडसस्टमों के डलए उपयुक्त 

बनाना था और ितशमान मामले में आांतररक जैसे पास रखना था। इस थीडसस में दो नए 

पुनर्नशमाशण एल्गोररदम डिकडसत ककए गए थे। सबसे पहल ेर्ास्त्रीय मोंटे-कालो (एमसी) तकनीक 

का डिस्तार "एमसी-आरजीए" तकनीक के रूप में जाना जाता था, डजसस ेएल्गोररदम की 

अडभसरण और कम्प्यूटेर्नल गडत को िास्तडिक कोडित जेनेरटक एल्गोररदम (आरजीए) का 

उपयोग करके बढाया जाता ह।ै इस सांर्ोधन के पररणामस्िरूप डिटेक्टर पैरामीटर अनुमान के 

डलए आिश्यक कम्प्यूटेर्नल समय में भारी कमी आई ह ैऔर पूरी तरह से "दरूी-गणना 

मानडचत्र" की आिश्यकता को समाप्त कर दतेा ह,ै जो पहल ेआरपीटी प्रयोग के डलए डनडहत 

था। आरपीटी आिेदन ठोस हीटर रॉि के कारण "छाया प्रभाि" के कारण सांिहनी उबलते प्रिाह 

प्रयोगों तक ही सीडमत था। "छाया प्रभाि" मोंटे कालो (एमसी) तकनीक व्यथश पर आधाररत 

र्ास्त्रीय डस्थडत पुनर्नशमाशण एल्गोररदम बनाता ह।ै इस अध्ययन में एक उपन्यास टै्रसर कण 

पुनर्नशमाशण एल्गोररदम आधाररत समथशन िेक्टर ररग्रेर्न (एसिीआर) और प्रासांडगकता िेक्टर 



मर्ीन (आरिीएम) लाग ूककया गया ह।ै मर्ीन पनुर्नशमाशण आधाररत पुनर्नशमाशण एल्गोररदम डस्थडत 

पुनर्नशमाशण के मामले में बहुत तेज़ और सटीक थे। 

औसत चरण िेग और अन्य "अर्ाांडत" मात्रा प्रदान करने के अलािा, आरपीटी ितशमान अध्ययन 

में तरल चरण की नकल करन ेिाल ेटै्रसर कण के िास्तडिक लग्राांडगयन प्रक्षेपिक्र को भी प्रदान 

करता ह।ै इसडलए, उबलत ेदो चरण प्रिाह में दखेी गई जरटल तरल गडतर्ीलता के डिस्तृत 

लक्षण के डलए यह एक उडचत डिडध ह।ै रेडियोधमी टै्रसर कण की डस्थडत समय श्ृांखला तरल 

गडत पर जानकारी सांदरे् दनेे की पनुरािृडि ्लॉट्स (आरपी) और पुनरािृडि मात्रा डिश्लेषण 

(आरक्यूए) द्वारा जाांच की गई थी।  
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